Author's response to reviews

Title: Hepatoprotective effects of Spirulina maxima in subjects with Non-alcoholic Fatty Liver Disease (NAFLD): three case reports.

Authors:

Aldo Ferreira-Hermosillo (aldofh@tutopia.com)
Patricia Victoria Torres-Duran (pavitodu@yahoo.com.mx)
Marco Antonio Juarez-Oropeza (majo_ya@yahoo.com.mx)

Version: 4 Date: 21 January 2010

Author's response to reviews: see over
Dear Team of JMCR,

Please find attached the revised manuscript entitled “Hepatoprotective effects of *Spirulina maxima* in subjects with Non-alcoholic Fatty Liver Disease (NAFLD): three case reports” by Aldo Ferreira-Hermosillo, Patricia V. Torres-Duran, and Marco A. Juarez-Oropeza.

Thank you for your comments, all of them have been included in the revised manuscript.

Point by point:

1. The authors in the article are exactly as they are in the submission system.
2. In the abstract we have included the following information: gender, age, ethnic origin and nationality of each patient.
3. We are sure that the abstract is identical both in the manuscript and the submission system.
4. Table 1 was placed at the end of the manuscript. Also, we edited vertical lines, they are not visible.

Thanking for your kind attention,
Dr. Patricia Victoria Torres-Duran